Author’s response to reviews

Title: Is publication important for annual performance assessment?

Authors:

Robin L Walker (walkr@ucalgary.ca)
Lindsay Sykes (lisykes@ucalgary.ca)
Brenda R Hemmelgarn (Brenda.Hemmelgarn@albertahealthservices.ca)
Hude Quan (hquan@ucalgary.ca)

Version: 2 Date: 22 October 2009

Author’s response to reviews:

Originally we sent this paper to BMCMedicine. They felt that our manuscript is ideally suited to the scope and readership of another of our journals, BMC Medical Education. Thus we are submitting the paper to your journal.